Author’s response to reviews

Title: Predictors of Provider-Initiated HIV Testing and Counseling refusal by outpatient department clients in Wolaita zone, Southern Ethiopia: A case control study

Authors:

Wolde Faltamo (woldiefacha@gmail.com)
Wondwosen Kassahun (wondkyt@yahoo.com)
Abdulhalik Workicho (abdulhalikw@gmail.com)

Version: 4 Date: 10 Jun 2016

Author’s response to reviews:

Response to editor:

(1) Move your 'Strength and limitations of the study' section to before you 'Conclusions' section in your manuscript.

# It was corrected as commented by editor.

(2) In your 'Availability of Data and Materials' section, we note that you state "We assure that the data-sets could be deposited in publicly available repositories whenever appropriate." As your data is of a sensitive nature and, we expect, confidential, we doubt this is the case. Please therefore remove this statement and indicate whether the corresponding author can be contacted for the raw data or whether you are unable to share this for confidentiality reasons.

# It was corrected as suggested by editor.